Author’s response to reviews

Title: Knowledge and practice of adolescent females about menstruation and menstruation hygiene visiting a public healthcare institute of Quetta, Pakistan

Authors:

Judy Michael (judy_michael@hotmail.com)
Qaiser Iqbal (dr.qaiseruob@gmail.com)
Sajjad Haider (dr.sajjadhaidar007@gmail.com)
Adnan Khalid (dr.adnankhalid66@gmail.com)
Naheed Haque (afihaque@gmail.com)
Rabia Ishaq (rabiashaq72@gmail.com)
Fahad Saleem (fahaduob@gmail.com)
Mohamed Azmi Hassali (azmihassali@gmail.com)
Mohammad Bashaar (dr.mbashaar@gmail.com)

Version: 3 Date: 29 Nov 2019

Author’s response to reviews:

To
Editor
BMC Women’s Health

Dear Sir/Madam

Subject:
Submission of a revised manuscript

Sir/Madam

It is for your information that the revised manuscript entitled “Knowledge and practice of adolescent females about menstruation and menstruation hygiene visiting a public healthcare institute of Quetta, Pakistan” is submitted for consideration.
We are thankful to the reviewers for their constructive comments. The response is added as a rebuttal with this submission.

Thank you for considering this paper for publication. We look forward to your reply.

Sincerely,

Mohammad Bashaar
Corresponding author
Executive Director/ Health Policy Analyst
SMART Afghan International Trainings & Consultancy, Kabul, Afghanistan
Email: dr.mbashaar@gmail.com; Tel.: +93789171199